Author's response to reviews

Title: Hepatitis C virus E2 protein involve in insulin resistance through an impairment of Akt/PKB and GSK3beta signaling in hepatocytes

Authors:

Ming-Ju Hsieh (kaross1006@hotmail.com)
Kuang-Ping Lan (lankping@gmail.com)
Hao-Yu Liu (a_how1218@yahoo.com.tw)
Xiao-Zong Zhang (barkley34@mail.chis.doh.gov.tw)
Yaw-Feng Lin (lyf@mail.npue.edu.tw)
Shun-Fa Yang (ysf@csmu.edu.tw)
Hui-Ling Chiou (hlchiou@csmu.edu.tw)

Version: 2 Date: 2 March 2012

Author's response to reviews: see over
Dear Editor:

Please find the enclosed manuscripts entitled “Hepatitis C virus E2 protein involve in insulin resistance through an impairment of Akt/PKB and GSK3β signaling in hepatocytes” by Hsieh et al., for you to consider publishing in *BMC Gastroenterology*. We here confirm that no manuscript with substantially the same content has been submitted or published elsewhere. All authors have read and approved this manuscript. This manuscript is prepared from the most recent results of our study. Please send all related correspondences directly to me via contact information listed below. Your kind assistance in evaluating this paper is highly appreciated.

Best regards,

Hui-Ling Chiou, PhD
Professor,
School of Medical Laboratory & Biotechnology,
Chung Shan Medical University,
Taichung, Taiwan
TEL: 886-4-24730022-12423
FAX: 886-4-23248147
Email: hlchiou@csmu.edu.tw